Author's response to reviews

Title: Making efforts to reduce the heterosexual transmission of HIV/STDs: The Global Fund AIDS Program in Shandong province, China.

Authors:

dianming Kang (dmkang@sohu.com)
Xiaorun Tao (taoxiaorun1963@163.com)
meizhen Liao (liaomz161@126.com)
jianzhuo Li (jianzhuoli@gmail.com)
nan Zhang (barbara044@163.com)
xiaoyan Zhu (zxysdams81@126.com)
xiaoguang sun (liaomz161@gmail.com)
bin Lin (hivjiance@126.com)
shengli Su (ssl29@126.com)
lianzheng Hao (hivjiance@126.com)

Version: 2 Date: 16 November 2012

Author's response to reviews: see over
November 9, 2012

RE: Manuscript: Making efforts to reduce the heterosexual transmission of HIV/STDs: The Global Fund AIDS Program in Shandong province, China.

Dear editor,

Enclosed for your consideration is an original article entitled Making efforts to reduce the heterosexual transmission of HIV/STDs: The Global Fund AIDS Program in Shandong province, China.

This manuscript is submitted to the BMC public health exclusively. No conflict of interest was involved in this study and the manuscript was approved by all the authors. We would like to recommend as peer-reviewers any of the following experts:

Sibylle Kristensen, DrPH, MPH, MSPH
Assistant Professor
UAB, Department of Epidemiology
1530 3rd Ave South, RPHB 227C
Birmingham, AL 35294-0022
Phone: 205-975-5793
FAX: 205-934-7154
Email: sibylle@uab.edu

Sharaf Shah, MD, MPH
Professor and Director, Institute of Infectious Diseases
Dow University of Health Sciences
52 D-S-Flats,G.O.R.Sindh, Bath Island Karachi, Pakistan
Phone: +92-21-5837467
Email: drsharafshah@yahoo.com or sharafshah@cyber.net.pk

Yan Xiao, MD, DrPH
Professor, National Center for AIDS Control and Prevention
China Center for Disease Control and Prevention
Room 502, No. 42 Dongjing Road
Xuanwu District, Beijing 100050, P. R. China
Phone: 010-63040391 010-63040299-601
Fax: 010-63040391
Email: xiaoyan@chinaaids.cn

For convenience, if you have any questions, please don’t hesitate to contact us +86-531-8267-9508 (Beijing time) or email at liaomz161@126.com.
Sincerely,

Meizhen Liao, MD, PhD
On behalf of all authors
Shandong Center for Disease Control and Prevention
Jinan 250012, Shandong Province, P.R. China
Tel: +86-186-1528-1775; Fax: +86-531-8267-9508
E-mail: liaomz161@126.com